Author’s response to reviews

Title: Randomized community trial on nosocomial infection control educational module for nurses in public hospitals in Yemen: a study protocol

Authors:

Gamil Alrubaiee (alrubaiee73@gmail.com)

Anisah Baharom (anisbaharom@yahoo.com;b_anisah@upm.edu.my)

Faisal Ibrahim (faisalupm@upm.edu.my)

Hayati Shahar (hayatik@upm.edu.my)

Shaffe Daud (shaffee@upm.edu.my)

Huda Basaleem (hudabasaleem92@yahoo.com)

Version: 1 Date: 06 Oct 2017

Author’s response to reviews:

Manuscript Title: Randomized community trail on nosocomial infection control educational module for nurses in public hospitals in Yemen: a study protocol.

Dear Editor,

We would like to submit our revised manuscript entitled above for publication in BMC Nursing, based on your valuable comments. We wish to confirm that this protocol currently on going and at the analysis stage and no publications containing the results of this study have already been published or submitted to any journal. Also, we wish to confirm that there are no known conflicts of interest associated with this publication and there has been no significant financial support for this work that could have influenced its outcome. All authors have contributed significantly, and have approved the manuscript and agree with submission to your esteemed journal. The requested ethical documentation have forwarded to BMCSeriesEditorial@biomedcentral.com.

Also we added all declaration sections to the manuscript based on your recommendations.
Thank you.

Yours Sincerely